Author's response to reviews

Title: Computer-aided transrectal ultrasound: Does prostate HistoScanningTM improve detection performance in repeat biopsies?

Authors:

Moritz F Hamann (moritz.hamann@uksh.de)
Claudius Hamann (claudius.hamann@uksh.de)
Arik Trettel (atrettel@yahoo.com)
Klaus-Peter Jünemann (Klaus-Peter.Juenemann@uksh.de)
Carsten M Naumann (CarstenMaik.Naumann@uksh.de)

Version: 5 Date: 29 June 2015

Author's response to reviews: see over
Dear editors,

On behalf of all authors, I am hereby sending you the revised manuscript (MS: 1000047284159681) with the title “Computer-aided transrectal ultrasound: Does prostate Histo-Scanning TM improve detection performance of prostate cancer in repeat biopsies”, Biomedcentral Urology.

We would like to thank the reviewers for their decision to accept the manuscript. With respect to the editorial requests, we have now made the changes accordingly.

New changes are marked in the text:

1. Ethical Approval/Permissions.
   We included the full name of the ethical committee in the Methods. (p. 4, ll. 90-91)

2. Abstract
   We included a sentence to describe the context of your study (why this research is being conducted), in addition to the study objectives. In accordance to the Instructions for Authors it was integrated in the background section of the abstract. (p. 2, ll.31-35)

We hope that the revised version of our manuscript conforms to the journal style and can now be accepted for publication in Biomedcentral Urology. We are pleased to have been given the opportunity to improve the manuscript and we appreciate the editors’ and the reviewers’ decision.

Yours sincerely

Moritz Franz Hamann, on behalf of all authors